Author’s response to reviews

Title: Perceptions and intentions toward medical assistance in dying among Canadian medical students

Authors:

James Falconer (james.m.falconer@gmail.com)
Félix Couture (felix.couture@mail.mcgill.ca)
Koray Demir (koray.demir@mail.mcgill.ca)
Michael Lang (michael.lang@mail.mcgill.ca)
Zachary Shefman (zachary.shefman@mail.mcgill.ca)
Mark Woo (mark.woo@mail.mcgill.ca)

Version: 1 Date: 02 Mar 2018

Author’s response to reviews:

Thank you for alerting us to these improvements. I will address each of them in turn:

1. We improved the Ethics Approval and Consent to Participate section to explain why only 6 ethics approvals were needed for data collection at 15 medical schools. The initial research ethics protocol was approved for data collection among all medical students at all medical schools in Canada. However, several institutions also required local REB approval for data collection in their Faculties of Medicine, hence the five additional REB approvals.

We also provided additional details in this section about how participant consent was obtained for this online survey, and indicated that the method of obtaining consent was approved by all REB protocols.

2. We added the email addresses for all authors following their institutional affiliations on the title page.
3. We corrected the email address of the corresponding author to ensure it is the same across the manuscript and submission system.

4. We renamed the abstract section "Objectives" to "Background" and we renamed the section "Introduction" to "Background" in the body of the manuscript.

5. We added a section "Legend of Figures", which appears following the references, and which lists the two tables that appear following the references, plus the one figure attached as a separate file.

Thank you for the opportunity to make these improvements. Please don't hesitate to get in touch for any additional improvements during the review process.

Best regards,

James Falconer